Author's response to reviews

Title: Obesity, Physical Activity, and the Urban Environment: Public Health Research Needs

Authors:

Russell P. Lopez (rptlopez@bu.edu)
H. Patricia Hynes (hph@bu.edu)

Version: 3 Date: 30 August 2006

Author's response to reviews:

Dear Dr. Grandjean:

We have retitled and revised our manuscript per your suggestions. Specifically:

1. We have changed the name of the article to "Obesity, Physical Activity, and the Urban Environment: Public Health Research Needs"
2. We shortened the article.
3. We eliminated many of the section headers and placed many of them into a table that illustrates the types of issues that affect urban communities.
4. We added to the abstract.
5. We have made other edits as requested.

Thank you for allowing us the opportunity to make these changes.

Russ Lopez